Author's response to reviews

Title: A theory of organizational readiness for change

Authors:

Bryan J Weiner (bryan weiner@unc.edu)

Version: 3 Date: 14 September 2009

Author's response to reviews:

Editors,

I've reviewed and accepted the changes recommended by your editorial staff. Please see the attached copy.